Author's response to reviews

Title: Mindful "Vitality In Practice": An Intervention to Improve the Work Engagement and Energy Balance among Workers; the development and design of the Randomised Controlled Trial

Authors:

Jantien van Berkel (j.vanberkel@vumc.nl)
Karin I Proper (ki.proper@vumc.nl)
Cécile RL Boot (crl.boot@vumc.nl)
Paulien M Bongers (paulien.bongers@tno.nl)
Allard J van der Beek (a.vanderbeek@vumc.nl)

Version: 2 Date: 29 August 2011

Author's response to reviews:

Dear Editor,

in reaction to the email we received, we would like to confirm that our study titled ‘Mindful "Vitality In Practice": An Intervention to Improve the Work Engagement and Energy Balance among Workers; the development and design of the Randomised Controlled Trial’, has been ethically approved and externally funded. Please find enclosed, the proofs of approval and funding. If you have any questions regarding these documents, please do not hesitate to contact us.

In addition to the above, we have included the Trial Registration Number the end of the Abstract in the submission system, as requested. Therefore, this revision has been made.

We look forward to hear your reaction to our manuscript.

Best regards,

Jantien van Berkel